Author’s response to reviews

Title: Assisting an Australian Aboriginal and Torres Strait Islander person with gambling problems: A Delphi study

Authors:

Kathy Bond (kathybond@mhfa.com.au)
Katrina Dart (katrinad@mhfa.com.au)
Anthony Jorm (ajorm@unimelb.edu.au)
Claire Kelly (clairek@mhfa.com.au)
Nicola Reavley (nreavley@unimelb.edu.au)
Kathy Bond (kathybond@mhfa.com.au)

Version: 1 Date: 20 Dec 2016

Author’s response to reviews:

Dear editor,

Thank you for the opportunity to revise this article before sending it to the reviewers. The suggested revisions have been made.